Author's response to reviews

Title: Therapeutic effect of the fruit hull of Gleditsia sinensis on lung inflammation in an LPS-induced acute lung injury model is associated with Nrf2

Authors:

Kyun Ha Kim (kyunhakim@pusan.ac.kr)
Min Jung Kwun (kmj1205@hanmail.net)
Chang Woo Han (yeast10@hanmail.net)
Ki-Tae Ha (hagis@pusan.ac.kr)
Jun-Yong Choi (orientdoct@gmail.com)
Myungsoo Joo (mjoo@pusan.ac.kr)

Version: 2 Date: 11 June 2014

Author's response to reviews:

Dear Reviewers,

We appreciate your careful reviews and comments, which help improve the quality of our manuscript. In this revision, we did our best to address the concerns of the reviewers. Changes made to the comments are highlighted red throughout the text of the manuscript. Attached are our point-by-point responses to the reviewer comments. We sincerely hope this revision adequately addresses the issues raised by the reviewers.

Sincerely yours,

Myungsoo Joo, Ph.D.